Author's response to reviews

Title: Evaluation of potential drug-herb interactions among a group of Palestinian patients with chronic diseases

Authors:

Rowa Al Ramahi (rawa_ramahi@najah.edu)
Nidal Jaradat (nidaljaradat@yahoo.com)
Rubaa Shalalfeh (laral_alkhatib@yahoo.com)
Sojoud Nasir (mutamed@yahoo.com)
Yazan Manasra (yazan-manasrah95@hotmail.com)
Ihab Shalalfeh (ihab_mm_sh@hotmail.com)
Yasmen Esam (yasmeen.banjabri@yahoo.com)

Version: 5 Date: 26 May 2015

Author's response to reviews:

Dear Editor,

Editor's comments:

The authors have addressed the issues raised by the reviewers however the work still requires copyediting for English language as there are a number of places in the text where there are grammatical and other language issues.

One of my colleagues whose first language is English has reviewed the manuscript. Her modifications are in red. Also, your kind reviewers have modified the language. I hope it is accepted now.